Author’s response to reviews

Title: Variable selection under multiple imputation using the bootstrap in a prognostic study

Authors:

Martijn W Heymans (mw.heymans@vumc.nl)
Stef van Buuren (s.vanbuuren@pg.tno.nl)
Dirk L Knol (d.knol@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Henrica CW de Vet (hcw.devet@vumc.nl)

Version: 5 Date: 2 May 2007

Author’s response to reviews: see over
3 May 2007

Dear editor,

We enclose our final response to the remarks of the reviewers point by point below before final publication of the manuscript.

Reviewer 1 (Royston)
Minor Essential Revisions

1. We replaced the reference on page 15 of Austin & Tu [13] into that of Sauerbrei & Schumacher [14].
2. We changed the word “barrel” into “channel” as suggested by this reviewer.

Reviewer 2 (Yang)

We fulfilled the only Discretionary Revision suggested by this reviewer and have proofread the manuscript.

We trust that our manuscript is now publishable.

Sincerely yours, on behalf of the co-authors,

Martijn W Heymans

VU University Medical Center
EMGO Institute
Van der Boechorststraat 7
1081 BT Amsterdam
The Netherlands
M: mw.heimans@vumc.nl